

City Research Online

City, University of London Institutional Repository

Citation: Fu, R., Kraft, A., Tian, X., Zhang, H. & Zuo, L. (2020). Financial Reporting Frequency and Corporate Innovation. The Journal of Law and Economics, 63(3), pp. 501-530. doi: 10.1086/708706

This is the accepted version of the paper.

This version of the publication may differ from the final published version.

Permanent repository link: https://openaccess.city.ac.uk/id/eprint/23498/

Link to published version: https://doi.org/10.1086/708706

Copyright: City Research Online aims to make research outputs of City, University of London available to a wider audience. Copyright and Moral Rights remain with the author(s) and/or copyright holders. URLs from City Research Online may be freely distributed and linked to.

Reuse: Copies of full items can be used for personal research or study, educational, or not-for-profit purposes without prior permission or charge. Provided that the authors, title and full bibliographic details are credited, a hyperlink and/or URL is given for the original metadata page and the content is not changed in any way.

 City Research Online:
 http://openaccess.city.ac.uk/
 publications@city.ac.uk

Financial Reporting Frequency and Corporate Innovation*

Renhui Fu <u>renhuifu@sjtu.edu.cn</u> Shanghai Jiao Tong University

Arthur Kraft <u>Arthur.Kraft.1@city.ac.uk</u> *City University London*

Xuan Tian <u>tianx@pbcsf.tsinghua.edu.cn</u> *Tsinghua University*

Huai Zhang <u>HuaiZhang@ntu.edu.sg</u> Nanyang Technological University

> Luo Zuo <u>luozuo@cornell.edu</u> *Cornell University*

December 27, 2019

ABSTRACT

We examine how the regulation of financial reporting frequency affects corporate innovation. We use a difference-in-differences approach based on a sample of treatment firms that experience a change in their reporting frequency and matched industry peers and control firms whose reporting frequency remains unchanged. We find that higher reporting frequency significantly reduces treatment firms' innovation output, but find no evidence that the net externality effect on industry peers is statistically significant. Together, our results are consistent with the hypothesis that frequent reporting induces managerial myopia and impedes corporate innovation.

Key words: regulation, reporting frequency, innovation, managerial myopia, externality.

JEL classification: G14, G18, M41, M45.

^{*} We appreciate the helpful comments of Dhammika Dharmapala (editor), an anonymous editor, an anonymous referee, Elizabeth Berger, Tim Gray, Yifei Mao, David Ng, and seminar participants at Bocconi University and Cornell University. We thank Fan Feng, Zhaojun Huang, Ashish Ochani, JunHyeog Oh, and Ying Zhao for outstanding research assistance. Fu acknowledges financial support from the China Ministry of Education Youth Foundation for Humanities and Social Sciences Research (15YJC630020) and the National Natural Science Foundation of China (71772119, 71972131). Tian acknowledges financial support from the National Natural Science Foundation of China (71790591, 71825002, 91746301). Zhang acknowledges financial support from the Clifford H. Whitcomb Faculty Fellowship. We remain responsible for any remaining errors or omissions.

Financial Reporting Frequency and Corporate Innovation

ABSTRACT

We examine how the regulation of financial reporting frequency affects corporate innovation. We use a difference-in-differences approach based on a sample of treatment firms that experience a change in their reporting frequency and matched industry peers and control firms whose reporting frequency remains unchanged. We find that higher reporting frequency significantly reduces treatment firms' innovation output, but find no evidence that the net externality effect on industry peers is statistically significant. Together, our results are consistent with the hypothesis that frequent reporting induces managerial myopia and impedes corporate innovation.

Key words: regulation, reporting frequency, innovation, managerial myopia, externality.

JEL classification: G14, G18, M41, M45.

1. Introduction

What drives corporate innovation, which is critical to both a nation's economic growth (Solow 1956, 1957; Romer 1990) and a firm's competitive advantage (Porter 1992)? A fastgrowing literature tackles this question, exploring empirical links between corporate innovation and a variety of firm-, industry-, and market-level characteristics (see He and Tian (2018) for a recent review). To stimulate innovation, governments typically implement policies providing protection of intellectual property rights. The economic consequences of these policies have been widely documented (e.g., Lerner 2009). In addition, recent research has studied whether and how the incentives to innovate are influenced by public policies not directly targeted at innovation, such as health policy (Finkelstein 2004), labor laws (Acharya, Baghai, and Subramanian 2013, 2014), bankruptcy codes (Acharya and Subramanian 2009; Cerqueiro, Hegde, Penas, and Seamans 2017), and tort laws (Galasso and Luo 2017). In this paper, we examine whether and how the regulation of reporting frequency affects corporate innovation. We study the effects of a change in reporting frequency on both the treatment firms that experience such a change and their industry peers whose reporting frequency remains unchanged as both effects are relevant from a regulator's perspective (Roychowdhury, Shroff, and Verdi 2019).

Motivating innovation is difficult for most firms. Different from routine tasks that rely on well-known approaches, corporate innovation entails the exploration of unknown methods that typically have a high probability of failure, involve multi-stage investment, and take years to generate positive returns (Holmstrom 1989). Therefore, to effectively motivate innovation, managers must be protected from external short-term pressure, and short-term failures must be tolerated (Manso 2011). Yet more frequent financial reporting likely intensifies short-term pressure from capital markets and puts managers in a position in which their failure to meet short-term earnings targets is less tolerated. Therefore, more frequent financial reporting could induce managers to focus on short-term earnings, rather than long-term firm value, resulting in less innovation.

This hypothesis is supported by both theoretical work (Gigler et al. 2014) and survey evidence (Graham, Harvey, and Rajgopal 2005) and is closely related to recent research on the relation between reporting frequency and capital expenditures. For example, Kraft, Vashishtha, and Venkatachalam (2018) find that firms listed on the U.S. stock exchanges decrease their capital investment levels, following a reporting frequency increase. However, Nallareddy, Pozen, and Rajgopal (2017) and Kajüter, Klassmann, and Nienhaus (2019) find no such evidence in the United Kingdom and Singapore. Our focus on corporate innovation differentiates our work from these studies in two important ways. First, unlike conventional investments (e.g., capital expenditures) which are initially capitalized and only gradually affect earnings via depreciation, corporate innovation is a long-term, risky, and idiosyncratic investment in intangible assets (Holmstrom 1989) and innovation expenditures (i.e., R&D) can have an immediate one-to-one negative effect on pre-tax earnings.¹ These features make innovation vulnerable to short-term pressure created by frequent reporting and well suited to testing theories of myopia. Second, we can measure both the quantity and quality of innovation output, based on patent information. Note that the relation between reporting frequency and innovation cannot be readily inferred from the mixed evidence on capital expenditures, as research shows that the same economic factor can have opposite impacts on these two types of investments.²

¹ Under current U.S. Generally Accepted Accounting Principles, research and development costs are expensed immediately. Under International Financial Reporting Standards, development costs can be capitalized as intangible assets.

 $^{^2}$ For example, using the same setting of brokerage closures and mergers to identify changes in analyst coverage, research documents drastically different effects of analyst coverage on capital expenditures and corporate innovation. While Derrien and Kecskes (2013) find that more analyst coverage leads to more capital expenditures (by reducing information asymmetry and the cost of capital), He and Tian (2013) find that more analyst coverage leads to a reduction in innovation (by imposing short-term pressure).

Although the above discussion highlights that increased frequency of financial reporting can hinder corporate innovation, the literature also suggests that more frequent reporting could lead to greater innovation for at least two reasons. More frequent reporting can improve firms' access to financing by lowering their cost of equity (Fu, Kraft, and Zhang 2012). A lower cost of equity helps relax a firm's financial constraints and allows it to invest more in innovation, which requires a significant amount of investment in both tangible and intangible assets. In addition, more frequent reporting could improve monitoring from capital markets and help discipline managers, who may be reluctant to invest in long-term projects. Moral hazard models suggest that managers who are not properly disciplined shirk or invest suboptimally in short-term projects that generate quicker and more certain returns (Grossman and Hart 1988; Harris and Raviv 1988, 1989). Frequent financial reporting exposes managers to more intensive monitoring by a variety of capital market players (such as financial analysts, short-sellers, and regulators) and motivates them to invest in long-term, value-enhancing projects.³ Given these tensions in the literature, the existence, direction, and economic magnitude of the effect of financial reporting frequency on corporate innovation are unresolved empirical questions.

Frequent reporting can also generate significant externalities for peer firms. On one hand, frequent reporting potentially reduces industry-level information asymmetry and helps industry peers identify investment opportunities or reduce agency frictions (e.g., Badertscher, Shroff, and White 2013; Shroff, Verdi, and Yu 2014; Shroff, Verdi, and Yost 2017; Arif and De Gorge 2019). This information spillover is likely to have a positive effect on industry peers' innovation. On the other hand, a firm's myopic behavior caused by frequent reporting can create short-term performance pressure on its industry peers and hinder their innovation.

³ Consistent with this disciplinary role of reporting frequency, Balakrishnan and Ertan (2018) find that greater reporting frequency is associated with an improvement in loan portfolio quality in the banking industry.

Therefore, the net externality effect of frequent reporting is unclear *ex-ante*.

We use the financial reporting frequency change in the U.S. as our empirical setting. The U.S. Securities and Exchange Commission (SEC) required annual financial reporting of listed firms in 1934, increased the frequency to semi-annual reporting in 1955, and further increased it to quarterly reporting in 1970. We perform two event studies to gauge the overall effect of reporting frequency on firm value. If more frequent reporting causes a firm's managers to become more myopic, the value of the firm would fall. Using a three-day event window around the SEC announcement of mandatory quarterly reporting (on September 15, 1969), we find a significant negative effect of 1% on the market value for firms that reported semi-annually but no significant effect for firms that already reported quarterly.⁴ These results suggest that quarterly reporting is net costly for these semi-annual reporters, which explains why they had not voluntarily reported this way previously.

While the mandate on quarterly reporting has been in effect for almost five decades, President Trump recently (on August 17, 2018) asked the SEC, via a tweet, to review quarterly reporting and reconsider semi-annual reporting for public companies.⁵ Using a three-day event window around Trump's tweet, we find a significant positive effect of 0.6% on the market value for firms in which innovation matters a lot but a relatively weaker effect of 0.3% on the market value for other firms. The significant difference in market reactions (i.e., 0.3%) between these two types of firms alleviates the concern that the positive market reaction for innovative firms reflects other implications of Trump's tweet (e.g., less burdensome disclosure requirements or more business-friendly regulation). Our results suggest that the cost of

⁴ Prior to 1970, many firms had already reported more frequently than required by the SEC, due to stock exchange listing requirements or pressure. As early as 1923, the NYSE required newly listed firms to provide quarterly reports and pressured already listed firms to do the same, and in 1926, it asked all listed firms to commit to quarterly reporting. The AMEX and other regional exchanges took similar actions in 1962. See more detailed descriptions in the work of Leftwich, Watts, and Zimmerman (1981) and Butler, Kraft, and Weiss (2007).

⁵ On August 17, 2018, President Trump tweeted "Stop quarterly reporting & go to a six month system." See the *Wall Street Journal* article by Michaels, Rapoport and Maloney (2018) for details.

quarterly reporting (i.e., exacerbating managerial myopia) matters more for innovative firms.

These two event studies provide preliminary evidence consistent with our hypothesis that frequent reporting induces managerial myopia and is net costly to innovative firms. To more directly test the link between financial reporting frequency and managerial myopia, we use observable innovation output to gauge the success of long-term investment in innovation, which is typically hard to observe and measure. Specifically, we construct three innovation output measures: the number of patent applications a firm files in a year that are eventually granted, the number of non-self-citations the firm's patents receive in subsequent years, and the economic value of patents, based on stock market reactions to patent grants (computed according to the method of Kogan et al. (2017)). These three measures capture patent quantity, quality, and economic value, respectively.

Our interim reporting frequency data are from the work of Butler, Kraft, and Weiss (2007) and Fu, Kraft, and Zhang (2012) and span the period 1951–1973. This empirical setting has three desirable features. First, there is substantial cross-sectional and time-series variation in firms' reporting frequency over this period. It is impossible to study the relation between reporting frequency and innovation using more recent U.S. data because almost all firms have followed the SEC's quarterly reporting requirement since 1970. Second, the SEC mandate affects only a subset of firms at a time, because some firms had already adopted more frequent reporting prior to the mandate due to stock exchange requirements or investor pressure. This feature allows us to observe plausible counterfactuals: what level of innovation productivity would firms have achieved in the absence of a reporting frequency change? The counterfactual is based on control firms with similar economic characteristics but that are not themselves subject to the reporting frequency change. Thus, we can use a difference-in-differences approach to tighten identification. Third, for this early sample period of 1951–1973, our patent data (ended in 2010) are unlikely to suffer from the usual truncation problems the innovation

literature has to deal with (Lerner and Seru 2017).

As a first step, we provide descriptive evidence on the trends of aggregate innovation in the economy. We plot the ratio of aggregate innovation by public firms, relative to aggregate innovation by other entities, as well as the individual and total trends over the sample period 1951–1973. We observe an overall upward trend for the ratio, suggesting that public firms' contribution to aggregate innovation generally increases over time. However, we observe a temporary decrease of public firm innovation around 1970 (when the SEC mandate on quarterly reporting took effect). This evidence suggests that the net impact of frequent reporting (aggregating both treatment effects and spillover effects on peer firms) on total innovation is negative. Interestingly, this temporary decrease in aggregate innovation by public firms is more than offset by an increase in aggregate innovation by other entities, leading to an increase in aggregate innovation. We fully acknowledge that these trends, while interesting, can only be interpreted as descriptive. Thus, we turn to firm-level analyses to strengthen empirical identification and provide tighter evidence.

We use a difference-in-differences approach to examine how regulation on financial reporting frequency affects corporate innovation. We designate firms that experience an increase in reporting frequency as treatment firms. We then use propensity-score matching to identify peer firms in the same industry and control firms in other industries (both with similar economic characteristics but whose reporting frequency remains unchanged). The treatment group is treated by increases in reporting frequency. The peer group is not subject to increases in reporting frequency but affected by the externality effect of increased reporting by firms in the treatment group. The control group is affected by neither increases in reporting frequency nor the externality effect.

In the difference-in-differences tests, we examine the effects of increases in reporting frequency on the innovation output of the treatment firms and their industry peers relative to the control firms. We find a significant reduction in innovation output for the treatment firms relative to the control firms. Specifically, the difference-in-differences estimators show that, compared with control firms, mandatory adopters (i.e., firms that increase their reporting frequency due to the SEC's requirement or exchange requirement) experience a decrease of 1.87 patents, 19.58 non-self-citations, and \$1.76 million worth of patent value after the mandatory switch. We find similar results for voluntary adopters (i.e., firms that increase their reporting frequency due to the demand of investors). These results suggest that frequent reporting induces managerial myopia and hinders innovation output, but this increase is not statistically different from that of the control firms. Our inferences are unchanged when we perform a difference-in-differences regression analysis with the matched sample or the full sample.

Overall, our evidence suggests that higher reporting frequency imposes short-term pressure on firm managers and impedes innovation, and we do not find evidence that the net externality effect on industry peers is statistically significant. These results could be of potential interest to regulators and policymakers in evaluating the costs and benefits of the quarterly reporting mandate.

The rest of the paper is organized as follows. Section 2 discusses the related literature and our contribution. Section 3 presents two event studies to gauge the overall effect of reporting frequency on firm value. Section 4 provides some descriptive evidence on the trends of aggregate innovation in the economy. Section 5 describes the sample selection, variable measurement, and summary statistics. Section 6 presents the main difference-in-differences results, and Section 7 concludes.

2. Related literature

Our contributions to the literature are threefold. First, our study adds a new angle to the literature on corporate innovation by identifying an important accounting practice, financial reporting frequency, as a crucial determinant of innovation. Studies have found that managerial incentives of investing in innovation are affected by various firm, industry, and market characteristics, including product market competition (Aghion et al. 2005), private equity ownership (Lerner, Sorensen, and Stromberg 2011), CEO overconfidence (Hirshleifer, Low, and Teoh 2012), institutional ownership (Aghion, Van Reenen, and Zingales 2013), financial analysts (He and Tian 2013), laws (Acharya and Subramanian 2009; Achara et al. 2013, 2014; Galasso and Luo 2017), market conditions (Nanda and Rhodes-Kropf 2013), corporate venture capitalists (Chemmanur et al. 2014), mergers and acquisitions (Bena and Li 2014), firm boundaries (Seru 2014), investors' attitudes toward failure (Tian and Wang 2014), banking competition (Cornaggia et al. 2015), bank interventions (Gu, Mao, and Tian 2018), and external financial dependence (Acharya and Xu 2017). While this line of work highlights many determinants of corporate innovation, the role of accounting practices has largely been ignored. Research in accounting typically focuses on the effect of a firm's financial reporting quality on its capital investment (e.g., Biddle and Hilary 2006; Biddle, Hilary, and Verdi 2009).⁶ A notable exception is the work of Zhong (2018), who documents that transparency enhances firm innovation in an international setting. We build on the theoretical work of Gigler et al. (2014) and provide empirical evidence that the frequency of financial reporting has a substantial effect on corporate innovation by exacerbating managerial myopia.

Second, our work contributes to the literature on financial reporting frequency. Research on the frequency of financial reporting largely focuses on its effects on firms'

⁶ See also Francis and Martin (2010); Bushman, Piotroski, and Smith (2011); Badertscher, Shroff, and White (2013); Balakrishnan, Core, and Verdi (2014); Goodman et al. (2014); Shroff, Verdi, and Yu (2014); Balakrishnan, Watts, and Zuo (2016), Garcia Lara, Garcia Osma, and Penalva (2016); and Shroff (2017, 2019).

information environments, such as the information content of annual reports (McNichols and Manegold 1983), earnings timeliness (Alford et al. 1993; Butler, Kraft, and Weiss 2007), and the cost of equity (Fu, Kraft, and Zhang 2012; Verdi 2012). Recent studies begin to examine the effects of frequent financial reporting on managerial decisions, such as investments in fixed assets (Nallareddy, Pozen and Rajgopal 2017; Kraft, Vashishtha and Venkatachalam 2018; Kajüter, Klassmann, and Nienhaus 2019), real activities manipulations (Ernstberger et al. 2017), cash holdings (Downar, Ernstberger and Link 2018), and banks' loan portfolio quality (Balakrishnan and Ertan 2018). Given the mixed evidence in the literature, Roychowdhury, Shroff, and Verdi (2019) conclude that whether an increase in reporting frequency decreases managers' investment horizon and induces myopia, or whether it increases transparency and serves a disciplinary role remains an open question. Our study sheds light on this important question by focusing on a firm's innovation, which is critical to a country's competitive advantages and long-term growth.

Moreover, we provide richer evidence on the economic consequences of reporting frequency than the literature in several ways. First, we conduct an event study to show a negative market reaction to the quarterly reporting mandate, which is consistent with firms incurring a firm-specific net cost and explains why firms do not voluntarily increase their reporting frequency. Second, an unexpected recent event, that is, President Trump's tweet, gives us the opportunity to demonstrate that the cost of quarterly reporting (i.e., exacerbating managerial myopia) outweighs its benefit (i.e., lowering the cost of equity), especially for innovative firms. Third, we complement our firm-level analyses with descriptive evidence on the trends of aggregate innovation in the economy. Fourth, we conduct separate analyses for the matched industry peers and assess on the externality of mandatory quarterly reporting. Understanding this spillover effect is important since one of the primary justifications for mandatory disclosure is externalities (see Minnis and Shroff (2017) for a thorough review).

Finally, our finding that more frequent reporting impedes corporate innovation is of interest to regulators and industry groups, who have recently debated on whether firms should be required to undertake more frequent interim financial reporting (e.g., Day 2003; European Commission 2004; Jopson 2006; Yiu 2009; Solomon 2011; Yahya 2016). The SEC is considering the pros and cons of replacing quarterly with semi-annual reporting (especially for smaller reporting companies).⁷ The United Kingdom started requiring firms to provide quarterly "Interim Management Statements" in 2007 but ended the requirement in 2014. To the extent that firms today face greater short-term pressure than in the past (Hersh 2016; Dimon and Buffett 2018; Stoll 2018), our results represent a lower-bound estimate of the impact of frequent financial reporting on corporate innovation.⁸

3. Event studies

We argue that more frequent reporting causes a firm's managers to become more myopic. If our argument is true, we would expect a drop in the value of a firm when it is required to report more frequently. Furthermore, we expect the effect to be more pronounced for firms where innovation plays an important role. We use event studies to test our expectations. The first event is the SEC's announcement of the quarterly reporting requirement on September 15, 1969. Our related results are reported in Panel A of Table 1.

[Insert Table 1 here.]

We gauge the market's reaction via CAR [0, 2], which is the cumulative abnormal return over the three-day window of [0, 2], with 0 being the event date. Its mean value is -1.0%, significant at the 10% level, for semiannual reporters (i.e., firms that reported semiannually before the announcement), while it averages 0.2%, statistically insignificant, for quarterly

⁷ <u>https://www.sec.gov/news/speech/international-developments-higgins.html</u>.

⁸ Given that the R&D expensing rules are different outside the United States, assessing the generalizability of our findings in an international setting is an interesting avenue for future research.

reporters (i.e., firms that reported quarterly prior to the announcement). The difference between the two types of firms is significant at the 5% level. This negative market reaction to the quarterly reporting mandate is consistent with firms incurring a firm-specific net cost and explains why firms might not have chosen to voluntarily increase reporting frequency.

The second event is President Trump's announcement on Twitter on August 17, 2018, which raised the possibility of dropping the quarterly reporting requirement. Panel B of Table 1 reports our results. We find that CAR [0, 2] is 0.6%, significant at the 1% level, for innovative firms (i.e., firms that have filed patents between 2005 and 2014), ⁹ while it is 0.3%, significant at the 1% level, for non-innovative firms (i.e., firms that have not filed patents between 2005 and 2014). The difference between the two types of firms is significant at the 5% level. Under the assumption that firms that file patents are the firms for which innovation matters, our results support the conjecture that the negative valuation impact of quarterly reporting is more severe for firms where innovation plays an important role.

Together, these two event studies provide preliminary evidence consistent with our hypothesis that frequent reporting induces managerial myopia and is net costly to innovative firms.

4. Descriptive evidence on aggregate trends

We hypothesize that high reporting frequency curbs innovation. As a first step, we provide descriptive evidence on the trends of aggregate innovation in the economy in Figure 1. The line "Total" represents the total number of patents filed in the year, divided by the total number of patents filed in 1973. From 1951 to 1965, the aggregate innovation in the United States increases steadily, reflecting the post-World War II prosperity and productivity gain.

⁹ Our patent data (collected from the U.S. Patent Office) end in 2014. To gauge a firm's innovativeness, we look at the most recent 10 years of a firm's patent filing history, i.e., 2005–2014.

From 1965 to 1969, it shows a declining trend. From 1969 to 1973, it increases again. Aggregate innovation is likely affected by geopolitics, macroeconomic conditions, and technological advances, in addition to reporting frequency.

Figure 1 also reports the proportion of patents generated by publicly listed firms and other entities through the lines labeled as "Public" and "Other," while the line "Relative" shows the number of patents filed by publicly listed firms divided by the number of patents filed by other entities.¹⁰ We find that, relative to other entities, the number of patents attributable to publicly listed firms increases between 1951 and 1968, decreases between 1968 and 1971, and increases from 1971 to 1973. The decrease between 1968 and 1971 is consistent with the conjecture that the quarterly reporting requirement damps the innovation of publicly listed firms (aggregating both treatment effects and spillover effects). We, however, acknowledge that these trends, while interesting, can only be interpreted as being descriptive.

[Insert Figure 1 here.]

5. Sample selection, variable measurement, and descriptive statistics

5.1.Sample selection

To strengthen empirical identification and provide tighter evidence, we turn to firmlevel analyses. Our sample is drawn from the work of Butler, Kraft, and Weiss (2007) and Fu, Kraft, and Zhang (2012), who hand-collect the data from *Moody's Industrial News Reports* covering the 1951–1973 period.¹¹ Reporting frequency is defined as one for annual reporters, two for semiannual reporters, three for firms reporting three times a year, and four for quarterly reporters. The following firms are excluded: firms not listed on either the NYSE or AMEX,

¹⁰ Other entities include private firms, universities, governments, and even individuals. Most of the patents are filed by firms.

¹¹ See Butler, Kraft, and Weiss (2007) and Fu, Kraft, and Zhang (2012) for more details on the data sources and composition of the original reporting frequency samples.

firms lacking CRSP or Compustat data, and firms in industries with distinctive disclosure requirements (e.g., utilities; financial services, insurance, and real estate firms; and railroad and other transportation companies). We merge this dataset with the innovation data downloaded from <u>http://iu.box.com/patents</u> (see Kogan et al. (2017) for a detailed description of the data). Following the innovation literature (e.g., He and Tian 2013), we set the innovation proxies to zero for firms without available patent or citation information. Our results are quantitatively similar if we drop the observations with missing innovation proxies. Our sample consists of 9,904 firm-year observations for the period from 1951 to 1973.

5.2. Innovation measures and control variables

We construct three measures to capture a firm's innovation output. The first is the number of patent applications a firm files in a year that are eventually granted (*PAT*). We use a patent's application year, instead of its grant year, because the application year arguably better captures the actual timing of innovation (Griliches, Pakes, and Hall 1988). A limitation of this measure is that it does not distinguish major innovations from marginal advances. To further gauge a patent's impact, we employ two other measures of corporate innovation output: the number of non-self-citations the firm's patents receive in subsequent years (*TCITE*) and the economic value of patents (*TSM*), based on stock market reactions to patent grants. The difference between these two measures is that the former mainly captures scientific impact, while the latter represents market value to firm shareholders. Our data for these innovation measures end in 2010. Since our sample period of 1951–1973 ends long before 2010, our patent variables are unlikely to suffer from the typical truncation problems the innovation literature needs to deal with.

Control variables include firm size, *LNMV*, measured by the natural logarithm of firm market capitalization; investment in innovation, *RD*, measured by R&D expenditures scaled

by total assets;¹² profitability, *ROA*, measured by return on assets; asset tangibility, *PPE*, measured by net property, plant, and equipment scaled by total assets; leverage, *LEV*, measured by total debt-to-total assets; investment in fixed assets, *CAPEX*, measured by capital expenditures scaled by total assets; product market competition, *HERF* measured by the Herfindahl index based on annual sales;¹³ growth opportunities, *Q*, measured by Tobin's Q; financial constraints, *HPINDEX*, a financial constraint measure based on firm size and age that is developed by Hadlock and Pierce (2010);¹⁴ firm age, *LNAGE*, measured by the natural logarithm of one plus the number of years the firm is listed on Compustat; and stock illiquidity, *AMIHUD*, measured by the yearly median of the Amihud (2002) price-impact measure (i.e., daily absolute stock return divided by the dollar trading volume measured in 1000s).

5.3.Descriptive statistics

Panel A of Table 2 reports the distribution of sample firms by reporting regimes. In the period of 1951–1954, when only annual reporting is required, 22.55% of our sample firms report semiannually, and 66.67% of them report quarterly. In the period of 1955–1969, when semiannual reporting starts to be compulsory, 86.61% of our sample firms report quarterly. In the period of 1970–1973, 96.60% of our sample firms report quarterly. These three periods account for 5.06%, 59.86%, and 35.08% of our sample, respectively. There are 1.68% of our sample observations that report three times a year because firms may decide to switch from semi-annual to quarterly reporting in the middle of the fiscal year. Our sample period provides both cross-sectional and time-series variation in reporting frequency and provides an ideal

¹² Our results are largely unchanged when we control for cumulative R&D expenditures over the current year and the previous one, two, or three years.

¹³ We also include in our regressions the squared Herfindahl index, *HERF_SQR*, to account for the nonlinear effect of product market competition (Aghion et al. 2005).

¹⁴ We do not use the more current measures of financial constraint based on textual analyses of 10-Ks (e.g., Hoberg and Maksimovic 2015; Buehlmaier and Whited 2018), because such measures are not available for our sample period.

setting for our investigation. In total, our full sample consists of 1,117 unique firms and 9,904 firm-year observations.¹⁵

[Insert Table 2 here.]

Panel B of Table 2 provides descriptive statistics for our full sample. *PAT* has a mean of 6.46 and a median of 0. The mean and median of *TCITE* are 46.26 and 0, respectively. *TSM* has a mean of 4.32, while its median is 0. *LNMV* has a mean of 3.83 and a median of 3.70. The mean and median of *RD* are 0.006 and 0, respectively. The mean of *ROA*, *PPE*, *LEV*, and *CAPEX* indicate that an average firm in our sample has a return on asset ratio of 14.8%, its *PPE* (net) is about 32% of its total assets, its leverage ratio is 22%, and its capital expenditure is about 6% of its total assets. The mean of *HERF* is 0.48, while the mean of *Q* is 1.7. The mean of *HPINDEX* is -2.37, and the median is -2.43. The mean of *LNAGE* is 1.46, while the average of *AMIHUD* is 0.015.

6. Difference-in-differences analyses

6.1.Treatment, peer, and control groups

We argue that financial reporting frequency affects economy-wide innovation in various ways: First, frequent reporting induces managerial myopia and hinders innovation for those reporting firms. Second, frequent reporting improves firms' access to financing and monitoring from capital markets and thus enhances firm innovation. Third, frequent reporting potentially reduces industry-level information asymmetry and this information spillover is likely to have a positive effect on industry peers' innovation. Fourth, a firm's myopic behavior caused by frequent reporting can create short-term performance pressure on its industry peers and hinder their innovation. To empirically assess the treatment and externality effects of

¹⁵ Due to some of our sample firms switching their reporting frequency during the sample period, the total number of unique firms in our full sample does not equal to the sum of the number of unique firms across different frequency.

frequent reporting on innovation, we divide the full sample into three groups: the treatment group (treated by increases in reporting frequency and possibly also by the externality effect from other firms in this group), the peer group (not subject to increases in reporting frequency but affected by the externality effect of increased reporting by firms in the treatment group), and the control group (affected by neither increases in reporting frequency nor the externality effect).

We construct the treatment, peer, and control groups of firms using propensity-score matching. Specifically, we first use the full sample to run an ordered probit model to estimate the propensity score related to the change of reporting frequency (see Appendix B). We then use the predicted probabilities, or propensity scores, to perform a nearest-neighbor propensity-score matching to identify peer firms and control firms. By construction, peer firms and control firms have characteristics similar to treatment firms, but their reporting frequency remains unchanged. We require peer firms to be in the same industry (based on Fama-French 48 industries) as treatment firms because externalities are most likely to occur among industry peers. Control firms are from other industries.

Table 3 reports the distributions of treatment firms (firms that increase their frequency) used in the tests. Panel A reports the distribution according to the change in reporting frequency. In the period of 1951–1954, 57 firms increase their reporting frequency to the semiannual level, 20 firms to three times a year, and 89 firms to the quarterly level. In the period of 1955–1969, 252 firms increase to the semiannual level, 157 firms to three times a year, and 426 firms increase to the quarterly level. In the period of 1970–1973, the number of firms increasing to the semiannual level, the three-times-a-year level, and the quarterly level is 9, 13 and 52, respectively. Our findings are largely consistent with those of Kraft et al. (2018).¹⁶

¹⁶ There are very few cases in which firms temporarily decrease their reporting frequency. We find few effects of these temporary reporting changes on firms' innovation output.

[Insert Table 3 here.]

Panel B presents the distribution of treatment firms, according to the reason for the change in reporting frequency. Firms may switch their reporting frequency because of the SEC's regulation, the stock exchange's requirements, or demand from investors. Specifically, we conclude that the switch is due to the SEC's regulation if the firm increased the frequency to the semiannual level starting in 1955 or increased the frequency to the quarterly level after 1967. The switch is deemed as a result of exchange requirement if the firm is listed on the AMEX and increased its frequency to the quarterly level starting one year before and up to two years after 1962 (the year in which the AMEX started urging existing firms and requiring newly listed firms to switch to quarterly reporting). During our sample period, there was no change in NYSE's listing rules regarding reporting frequency. We assume that, if firms are not required by either the SEC or the stock exchange to switch their reporting frequency, the reporting frequency switches are due to the demand of investors. Overall, there are 366 switches as a result of the SEC regulation, 133 switches as a result of exchange requirements, and 576 switches as a result of investor demand.

Our matched sample includes firm-year observations for treatment, peer, control firms over a six-year window centered on the year of the switch in the reporting frequency (Fang, Tian, and Tice 2014). Out of the 1,075 treatment firms with reporting frequency increases, 491 firms engage in some patenting activities during the sample period.¹⁷

Following Kraft et al. (2018), we classify firms that increase their reporting frequency due to the SEC's requirement or exchange requirement as mandatory increasers and all others as voluntary increasers. In total, 499 firms experience a mandatory increase in reporting frequency, and 576 firms experience a voluntary increase in reporting frequency. The existence

¹⁷ For expositional simplicity, the number of treatment firms here refers to the number of unique treatments (not the number of unique firms).

of both mandatory and voluntary adopters suggests that the cost-benefit tradeoff varies across firms. Firms voluntarily adopt more frequent reporting when the benefit of doing so (e.g., lower cost of equity) outweighs the cost (e.g., reduced innovation), while the opposite is likely true for mandatory adopters.

We depict in Figure 2 the values of the three innovation measures for the six years surrounding the mandatory increase in reporting frequency for the treatment, peer, and control groups. Year 0 (omitted from the figures) is the year of the switch. Panels A to C show that the two lines representing innovation output for the treatment group and control group trend closely in parallel in the three years leading up to the mandatory increase in reporting frequency. After the increase, the two lines start to diverge: innovation output increases slightly for the control firms, and it drops substantially for the treatment firms. The two lines representing innovation output for the peer group and control group trend closely in parallel over the six-year window. If we use peer firms as the benchmark, we reach the same conclusion that increases in reporting frequency in support of the parallel-trend assumption underlying the difference-in-differences analysis. It also shows that a mandatory increase in reporting frequency results in a lower level of innovation output for the treatment group but the net externality effect on the peer group seems limited.

[Insert Figure 2 here.]

6.2. Simple difference-in-differences tests

We use a difference-in-differences approach and compare the innovation output of treatment firms or their industry peers to that of comparable control firms. The difference-indifferences approach has three key advantages. First, it alleviates the concern that the timeseries trend, rather than a change in reporting frequency, drives the change in innovation output. Second, we can conduct tests for firms that change their reporting frequency as a result of the SEC or stock exchange mandate (rather than a firm's choice).¹⁸ Lastly, the difference-indifferences approach controls for unobserved constant differences between the treatment (or peer) group and the control group.

Based on the matched sample, we first conduct univariate tests to obtain the differencein-differences estimators. We adjust the innovation proxies by the average values for each year to remove aggregate time trends. Table 4 presents the results. We separately examine mandatory and voluntary changes, because mandatory changes in reporting frequency are unlikely driven by an individual firm's choice and hence provide better identification. We report the results for mandatory increases in Panel A and voluntary increases in Panel B. In both panels of Table 4, row (1) reports the change in innovation activities for treatment firms after the switch in financial reporting frequency. Specifically, we report the average change in the number of patents (PAT), the average change in the number of non-self-citations (TCITE), and the average change in the economic value of patents (TSM). These measures are computed by first subtracting patent counts/citation counts/economic values over the three years preceding the switch in reporting frequency from the patent counts/citation counts/economic values over the three years following the switch in reporting frequency for each treatment firm. The differences are then averaged over the treatment group. Similarly, we calculate the average change in patent counts/citation counts/economic values for the peer and control groups and report them in rows (2) and (3). In rows (4) to (6), we report the mean difference-in-differences (DiD) estimators and the corresponding two-tailed *t*-statistics, testing the null hypothesis that the estimators are zero.

[Insert Table 4 here.]

¹⁸ A caveat is that our sample does not include firms that choose to delist in the presence of additional disclosure mandates (e.g., Bushee and Leuz 2005).

Panel A of Table 4 reports the results for mandatory increases in reporting frequency. We find that these treatment firms experience a significant decrease in innovation output, consistent with our hypothesis that more frequent reporting leads to less corporate innovation; by contrast, the peer firms experience a significant increase in innovation output and there is no significant change for control firms. The difference-in-differences estimators in row (4) suggest that, on average, a mandatory increase in reporting frequency results in a decrease of 1.87 patents, 19.58 non-self-citations and \$1.76 million in economic value for the treatment firms relative to the control firms. We find similar treatment effects when using peer firms as the benchmark group (see row (6)).

We also assess whether increases in reporting frequency affect peer firms in the industry. Externalities influence peer firms in two ways. On one hand, a mandatory increase in reporting frequency of treatment firms reduces industry-level information asymmetry and thereby encourages innovation of peer firms. On the other hand, it elevates the short-termism of treatment firms and, through peer pressure, imposes similar changes on peer firms, resulting in lower innovation. Results in row (5) show that, on average, the net effect of externalities is statistically insignificant. This insignificant result should be interpreted with caution as it may reflect that our tests on externalities lack statistical power and our setting is not conducive for detecting externality effects. To further examine the economic significance (or size) of the externality effect for peer firms relative to control firms, we construct the 95% confidence intervals based on the estimates in row (5), which are [-1.76, 0.45] for patents, [-14.06, 17.57] for non-self-citations, and [-0.95, 1.43] for the economic value. Thus, the largest externality effect that can be ruled out at the 95% level appears to be relatively big and comparable to the average treatment effects estimated in row (4), though we observe both large positive and negative externalities.

Panel B of Table 4 reports the results for voluntary increases in reporting frequency. We find that the innovation output of the treatment firms decreases after the switches (row (1)). We find all three innovation measures increase significantly after the switch for peer firms and, to a less extent, for control firms (rows (2) and (3)). The magnitude of the difference-indifferences estimators suggests that, on average, a voluntary increase in reporting frequency results in a decrease of 2.15 patents, 12.25 non-self-citations, and \$2.25 million in economic value of patents compared with the control firms. Our results are similar when we use peer firms as the benchmark. The net effect of externality is statistically insignificant.¹⁹

6.3. Difference-in-differences regression analyses

In this subsection, we use the matched sample to conduct difference-in-differences regression analyses to obtain our main results. Specifically, following Fang, Tian, and Tice (2014), we use firm-year observations for treatment, peer, and control firms over a six-year window centered on the year of the switch in the reporting frequency, and estimate the following model:

 $INNOV = \alpha + \beta_{1}TREAT \times BEFORE^{2} + \beta_{2}TREAT \times BEFORE^{1} + \beta_{3}TREAT \times AFTER^{1}$ $+ \beta_{4}TREAT \times AFTER^{2} + \beta_{5}TREAT \times AFTER^{3} + \beta_{6}PEER \times BEFORE^{2}$ $+ \beta_{7}PEER \times BEFORE^{1} + \beta_{8}PEER \times AFTER^{1} + \beta_{9}PEER \times AFTER^{2} + \beta_{10}PEER \times AFTER^{3}$ $+ \beta_{11}BEFORE^{2} + \beta_{12}BEFORE^{1} + \beta_{13}AFTER^{1} + \beta_{14}AFTER^{2} + \beta_{15}AFTER^{3}$ $+ Firm Fixed Effects + Year Fixed Effects + \varepsilon.$ (1)

The dependent variable is one of the three innovation output measures (*PAT*, *TCITE*, and *TSM*). *TREAT* is a dummy variable that equals one for treatment firms and zero otherwise; *PEER* is a dummy variable that equals one for industry peers and zero otherwise; *BEFORE*² is

¹⁹ The largest externality effect that can be ruled out at the 95% level still appears to be relatively big. The 95% confidence intervals based on the estimates in row (5) are [-1.73, 1.42] for patents, [-5.59, 14.79] for non-self-citations, and [-1.91, 1.43] for the economic value.

a dummy variable that equals one if a firm-year observation is from the second year before the switch in reporting frequency (year -2) and zero otherwise; *BEFORE*¹ is a dummy variable that equals one if a firm-year observation is from the year before the switch in reporting frequency (year -1) and zero otherwise; *AFTER*¹ is a dummy variable that equals one if a firm-year observation is from the year immediately after the reporting frequency switch (year 1) and zero otherwise; *AFTER*² is a dummy variable that equals one if a firm-year observation is from the year immediately after the reporting frequency switch (year 1) and zero otherwise; *AFTER*² is a dummy variable that equals one if a firm-year observation is from the second year after the switch (year 2) and zero otherwise; *AFTER*³ is a dummy variable that equals one if a firm-year observation is from the third year after the switch (year 3) and zero otherwise. Also, we include the firm and year fixed effects.

The key coefficient estimates are β_1 to β_{10} . A statistically insignificant coefficient estimate of β_1 , β_2 , β_6 , or β_7 suggests that the parallel-trend assumption is not violated. Negative and significant coefficient estimates of β_3 , β_4 , or β_5 suggest that, compared with control firms, treatment firms generate a smaller number of patents, patents with fewer citations, and patents with smaller economic value, in the years following the reporting frequency change. Significant coefficient estimates of β_8 , β_9 , or β_{10} suggest that reporting frequency increases generate a statistically significant externality on industry peers (relative to control firms).

[Insert Table 5 here.]

We report the regression results from estimating equation (1) in Table 5. Results related to the dependent variable *PAT*, *TCITE*, and *TSM* are reported in columns (1), (2), and (3). Panel A reports the results for mandatory increases in reporting frequency. β_1 , β_2 , β_6 , and β_7 are all statistically insignificant in columns (1) to (3), suggesting that the parallel-trend assumption is not violated. β_3 , β_4 , and β_5 are negative and significant in eight out of nine specifications, consistent with our hypothesis that more frequent reporting leads to less corporate innovation for the treatment firms. β_8 , β_9 , and β_{10} are all statistically insignificant. Panel B reports the results for voluntary increases in reporting frequency. Our results resemble those reported in Panel A. In all three columns, β_1 , β_2 , β_6 , and β_7 are all statistically insignificant, suggesting that the parallel-trend assumption is not violated; β_3 , β_4 , and β_5 are negative and significant in eight out of nine specifications, suggesting that compared with control firms, treatment firms experience a drop in innovation output. β_8 , β_9 , and β_{10} are again all statistically insignificant.

Overall, these findings are consistent with our univariate difference-in-differences estimator findings and suggest that increases in reporting frequency lead to drops in innovation output.

6.4. Full sample analysis

Following prior research (e.g., Fu, Kraft, and Zhang 2012; Fang, Tian, and Tice 2014; Kraft, Vashishtha and Venkatachalam 2018), our previous analyses are based on matched samples over a six-year window centered on the year of the switch in the reporting frequency. An advantage of this approach is that it allows us to identify the three groups of firms (i.e., treatment, peer, and control firms) around the relatively short event window and study both firm-level effects and spillover effects on industry peers. Separating the sample into the three groups over the full sample period 1951–1973 is not feasible as most industries included in our sample are "treated" over this period.²⁰ In this section, we use the full sample and a generalized different-in-differences estimator that exploits the staggered nature of the treatment effects as a robustness check. Specifically, we use firm-year observations for the full sample and estimate the following model:

²⁰ An industry is "treated" when at least one firm in this industry switches reporting frequency.

 $INNOV = \alpha + \beta_1 QUARTERLY \times POST_Q + \beta_2 SEMIANNUAL \times POST_S + CONTROLS$

+ Firm Fixed Effects + Year Fixed Effects + ε . (2)

The dependent variable is one of the three innovation output measures (*PAT*, *TCITE*, and *TSM*). *QUARTERLY* is a dummy variable that equals one for treatment firms that increase reporting frequency to the quarterly level and zero otherwise; *SEMIANNUAL* is a dummy variable that equals one for treatment firms that increase reporting frequency to the semiannual level and zero otherwise; *POST_Q* is a dummy variable that equals one if a firm-year observation is from a year after the reporting frequency switch to the quarterly level and zero otherwise; *POST_S* is a dummy variable that equals one if a firm-year observation is from a year after the reporting frequency switch to the semiannual level and zero otherwise. We include the standard set of control variables as in Fang, Tian, and Tice (2014) and the firm and year fixed effects.

The key coefficient estimates are β_1 and β_2 . A negative and significant coefficient estimate of β_1 (or β_2) suggests that, compared with control firms, treatment firms generate a smaller number of patents, patents with fewer citations, and patents with smaller economic value, in the years following the reporting frequency change to the quarterly (or semiannual) level.

[Insert Table 6 here.]

We report the regression results from estimating equation (2) in Table 6. Results related to the dependent variable *PAT*, *TCITE*, and *TSM* are reported in columns (1), (2), and (3). β_1 is negative and significant in all three specifications, consistent with our hypothesis that more frequent reporting leads to less corporate innovation for the treatment firms. The negative but statistically insignificant β_2 suggests that switching from annual reporting to semiannual reporting is not particularly costly to treatment firms. But this latter result should be interpreted with caution given the limited number of treatment firms that switch to semiannual reporting over the sample period (see Table 3).

To ensure that the treatment effects of quarterly reporting documented in Table 6 are not driven by differential pre-trends, we add leads (i.e., $BEFORE^2$ and $BEFORE^1$) and lags (i.e., $AFTER^0$, $AFTER^1$, $AFTER^2$, $AFTER^3$, and $AFTER^{4+}$) as interaction terms to the model as in Autor (2003). We also add industry-specific linear trends in the specification.²¹ Table 7 reports the results. In all three columns, the coefficient estimates on the lead variables are all statistically insignificant, suggesting that the parallel-trend assumption is not violated; the coefficient estimates on the lag variables are negative and significant in 12 out of 15 specifications, suggesting that compared with control firms, treatment firms experience a drop in innovation output.

[Insert Table 7 here.]

Overall, these findings are consistent with our matched sample results and suggest that increases in reporting frequency lead to drops in innovation output.

7. Conclusion

We provide empirical evidence on the effect of financial reporting frequency regulation on corporate innovation. Based on two events—the SEC announcement of the quarterly reporting requirement and President Trump's tweet on reconsidering semi-annual reporting our analyses suggest that frequent reporting induces managerial myopia and is net costly to innovative firms. We also observe a temporary decrease of public firm innovation around 1970 (when the SEC mandate on quarterly reporting became effective). Using a difference-indifferences design, we find that firms experiencing an increase in reporting frequency exhibit

²¹ We do not include firm-specific linear trends because doing so significantly reduces the power of the test due to the limited number of firm-year observations (relative to the number of firms).

a lower level of innovation output relative to control firms. We find no evidence that the net externality effect on industry peers is statistically significant. Overall, our results suggest that higher reporting frequency imposes short-term pressure on firm managers and hence impedes innovation. Our evidence shows the real consequences of interim reporting frequency and has important policy implications for regulators and firms.

References

- Acharya, Viral, Ramin Baghal, and Krishnamurthy Subramanian. 2013. Labor Laws and Innovation. *Journal of Law and Economics* 56: 997–1037.
- Acharya, Viral, Ramin Baghal, and Krishnamurthy Subramanian.2014. Wrongful Discharge Laws and Innovation. *Review of Financial Studies* 27: 301–346.
- Acharya, Viral, and Krishnamurthy Subramanian. 2009. Bankruptcy Codes and Innovation. *Review of Financial Studies* 22: 4949–4988.
- Acharya, Viral, and Zhaoxia Xu. 2017. Financial Dependence and Innovation: The Case of Public versus Private Firms. *Journal of Financial Economics* 124: 223–243.
- Aghion, Philippe, Nick Bloom, Richard Blundell, Rachel Griffith, and Peter Howitt. 2005. Competition and Innovation: An Inverted U Relationship. *Quarterly Journal of Economics* 120: 701–728.
- Aghion, Philippe, John Van Reenen, and Luigi Zingales. 2013. Innovation and Institutional Ownership. *American Economic Review* 103: 277–304.
- Alford, Andrew, Jennifer Jones, Richard Leftwich, and Mark Zmijewski. 1993. The Relative Informativeness of Accounting Disclosures in Different Countries. *Journal of Accounting Research* 31: 183–223.
- Amihud, Yakov. 2002. Illiquidity and Stock Returns: Cross-section and Time Series Effects. *Journal of Financial Markets* 5: 31–56.
- Arif, Salman, and Emmanuel T. De George. 2019. The Dark Side of Low Financial Reporting Frequency: Investors' Reliance on Alternative Sources of Earnings News and Excessive Information Spillovers. Indiana University Working Paper No. 17-7.
- Asker, John, Joan Farre-Mensa, and Alexander Ljungqvist. 2015. Corporate Investment and Stock Market Listing: A Puzzle? *Review of Financial Studies* 28: 342–390.
- Autor, David H. 2003. Outsourcing at Will: The Contribution of Unjust Dismissal Doctrine to the Growth of Employment Outsourcing. *Journal of Labor Economics* 21: 1–42.
- Badertscher, Brad, Nemit Shroff, and Hal D. White. 2013. Externalities of Public Firm Presence: Evidence from Private Firms' Investment Decisions. *Journal of Financial Economics* 109: 682–706.
- Balakrishnan, Karthik, John Core, and Rodrigo S. Verdi. 2014. The Relation between Reporting Quality and Financing and Investment: Evidence from Shocks to Financing Capacity. *Journal of Accounting Research* 52: 1–36.
- Balakrishnan, Karthik, and Aytekin Ertan. 2018. Banks' Financial Reporting Frequency and Asset Quality. *The Accounting Review* 93: 1–24.
- Balakrishnan, Karthik, Ross L. Watts, and Luo Zuo. 2016. The Effect of Accounting Conservatism on Corporate Investment during the Global Financial Crisis. *Journal of Business Finance and Accounting* 43: 513–542.
- Bena, Jan, and Kai Li. 2014. Corporate Innovations and Mergers and Acquisitions. *Journal of Finance* 69: 1923–1960.
- Biddle, Gary C., and Gilles Hilary. 2006. Accounting Quality and Firm-level Capital Investment. *The Accounting Review* 81: 963–982.
- Biddle, Gary C., Gilles Hilary, and Rodrigo S. Verdi. 2009. How Does Financial Reporting Quality Relate to Investment Efficiency? *Journal of Accounting and Economics* 48: 112– 131.
- Buehlmaier, Matthias, and Toni M. Whited. 2018. Are Financial Constraints Priced? Evidence from Textual Analysis. *Review of Financial Studies* 31: 2693–2728.
- Bushee, Brian J., and Christian Leuz. 2005. Economic Consequences of SEC Disclosure Regulation: Evidence from the OTC Bulletin Board. *Journal of Accounting and*

Economics 39: 233–264.

- Bushman, Robert M., Joseph D. Piotroski, and Abbie J. Smith. 2011. Capital Allocation and Timely Accounting Recognition of Economic Losses. *Journal of Business Finance and Accounting* 38: 1–33.
- Butler, Marty, Arthur G. Kraft, and Ira S. Weiss. 2007. The Effect of Reporting Frequency on the Timeliness of Earnings: The Cases of Voluntary and Mandatory Interim Reports. *Journal of Accounting and Economics* 43: 181–217.
- Cerqueiro, Geraldo, Deepak Hegde, María F. Penas, and Robert Seamans. 2017. Debtor Rights, Credit Supply, and Innovation. *Management Science* 63: 3311–3327.
- Chemmanur, Thomas, Elena Loutskina, and Xuan Tian. 2014. Corporate Venture Capital, Value Creation, and Innovation. *Review of Financial Studies* 27: 2434–2473.
- Cornaggia, Jess, Yifei Mao, Xuan Tian, and Brian Wolfe. 2015. Does Banking Competition Affect Innovation? *Journal of Financial Economics* 115: 189–209.
- Day, Phillip. 2003. Analysts and Regulators Debate Usefulness of Quarterly Reports. *The Wall Street Journal*, February 19.
- Derrien, François, and Ambrus Kecskes. 2013. The Real Effects of Financial Shocks: Evidence from Exogenous Changes in Analyst Coverage. *Journal of Finance* 68: 1407–1440.
- Dimon, Jamie, and Warren E. Buffett. 2018. Short-Termism Is Harming the Economy. *The Wall Street Journal*, June 6.
- Downar, Benedikt, Jürgen Ernstberger, and Benedikt Link. 2018. The Monitoring Effect of More Frequent Disclosure. *Contemporary Accounting Research* 35: 2058–2081.
- Ernstberger, Jürgen, Benedikt Link, Michael Stich, and Oliver Vogler. 2017. The Real Effects of Mandatory Quarterly Reporting. *The Accounting Review* 92: 33–60.
- European Commission. 2004. Directive 2004/109/EC of the European Parliament and of the Council. *Official Journal of the European Union* L 390: 38–57.
- Fang, Vivian W., Xuan Tian, and Sheri Tice. 2014. Does Stock Liquidity Enhance or Impede Firm Innovation? *Journal of Finance* 69: 2085–2125.
- Finkelstein, Amy. 2004. Static and Dynamic Effects of Health Policy: Evidence from the Vaccine Industry. *Quarterly Journal of Economics* 119: 527–564.
- Francis, Jere R., and Xiumin Martin. 2010. Acquisition Profitability and Timely Loss Recognition. *Journal of Accounting and Economics* 49: 161–178.
- Fu, Renhui, Arthur G. Kraft, and Huai Zhang. 2012. Financial Reporting Frequency, Information Asymmetry, and the Cost of Equity. *Journal of Accounting and Economics* 54: 132–49.
- Galasso, Alberto, and Hong Luo. 2017. Tort Reform and Innovation. *Journal of Law and Economics* 60: 385–412.
- Garcia Lara, Juan M., Beatriz Garcia Osma, and Fernando Penalva. 2016. Accounting Conservatism and Firm Investment Efficiency. *Journal of Accounting and Economics* 61: 221–238.
- Gigler, Frank, Chandra S. Kanodia, Haresh Sapra, and Raghu Venugopalan. 2014. How Frequent Financial Reporting Can Cause Managerial Short-Termism: An Analysis of the Costs and Benefits of Increasing Reporting Frequency. *Journal of Accounting Research* 52: 357–387.
- Goodman, Theodore H., Monica I. Neamtiu, Nemit Shroff, and Hal D. White. 2014. Management Forecast Quality and Capital Investment Decisions. *The Accounting Review* 89: 331–365.
- Graham, John R., Campbell R. Harvey, and Shiva Rajgopal. 2005. The Economic Implications of Corporate Financial Reporting. *Journal of Accounting and Economics* 40: 3–73.
- Griliches, Zvi, Ariel Pakes, and Bronwyn H. Hall. 1988. The Value of Patents as Indicators of Inventive Activity. NBER Working Paper No. 2083.

- Grossman, Sanford J., and Oliver D. Hart. 1988. One Share/One Vote and the Market for Corporate Control. *Journal of Financial Economics* 20: 175–202.
- Gu, Yuqi, Connie X. Mao, and Xuan Tian. 2018. Bank Interventions and Firm Innovation: Evidence from Debt Covenant Violations. *Journal of Law and Economics* 60: 637–671.
- Hadlock, Charles J., and Joshua R. Pierce. 2010. New Evidence on Measuring Financial Constraints: Moving Beyond the KZ Index. *Review of Financial Studies* 23: 1909–1940.
- Harris, Milton, and Artur Raviv. 1988. Corporate Control Contests and Capital Structure. *Journal of Financial Economics* 20: 55–86.
- Harris, Milton, and Artur Raviv.1989. The Design of Securities. Journal of Financial Economics 24: 255–287.
- He, Jie, and Xuan Tian. 2013. The Dark Side of Analyst Coverage: The Case of Innovation. *Journal of Financial Economics* 109: 856–878.
- He, Jie, and Xuan Tian. 2018. Finance and Corporate Innovation: A Survey. Asia-Pacific Journal of Financial Studies 47: 165–212.
- Hersh, David. 2016. Stop Letting Quarterly Numbers Dictate Your Strategy. *Harvard Business Review*, December 13.
- Hirshleifer, David, Angie Low, and Siew H. Teoh. 2012. Are Overconfident CEOs Better Innovators? *Journal of Finance* 67: 1457–1498.
- Hoberg, Gerard, and Vojislav Maksimovic. 2015. Redefining Financial Constraints: A Text-Based Analysis. *Review of Financial Studies* 28: 1312–1352.
- Holmstrom, Bengt. 1989. Agency Costs and Innovation. *Journal of Economic Behavior and Organization* 12: 305–327.
- Jopson, Barney. 2006. Big Four in Call for Real-time Accounts. *Financial Times*, November 8.
- Kajüter, Peter, Florian Klassmann, and Martin Nienhaus. 2019. The Effect of Mandatory Quarterly Reporting on Firm Value. *The Accounting Review* 94: 251-277.
- Kogan, Leonid, Dimitris Papanikolaou, Amit Seru, and Noah Stoffman. 2017. Technological Innovation, Resource Allocation and Growth. *Quarterly Journal of Economics* 132: 665– 712.
- Kraft, Arthur G., Rahul Vashishtha, and Mohan Venkatachalam. 2018. Frequent Financial Reporting and Managerial Myopia. *The Accounting Review* 93: 249–275.
- Leftwich, Richard W., Ross L. Watts, and Jerold L. Zimmerman. 1981. Voluntary Corporate Disclosure: The Case of Interim Reporting. *Journal of Accounting Research* 19: 50–77.
- Lerner, Josh. 2009. The Empirical Impact of Intellectual Property Rights on Innovation: Puzzles and Clues. *American Economic Review* 99: 343–348.
- Lerner, Josh, and Amit Seru. 2017. The Use and Misuse of Patent Data: Issues for Corporate Finance and Beyond. Harvard Business School Working Paper 18-042.
- Lerner, Josh, Morten Sorensen, and Per Stromberg. 2011. Private Equity and Long-run Investment: The Case of Innovation. *Journal of Finance* 66: 445–477.
- Manso, Gustavo. 2011. Motivating Innovation. Journal of Finance 66: 1823–1860.
- Mcnichols, Maureen, and James G. Manegold. 1983. The Effect of the Information Environment on the Relationship between Financial Disclosure and Security Price Variability. *Journal of Accounting and Economics* 5: 49–74.
- Michaels, Dave, and Michael Rapoport, and Jennifer Maloney. 2018. Trump Directs SEC to Study Six-Month Reporting for Public Companies. *The Wall Street Journal*, August 17.
- Minnis, Michael, and Nemit Shroff. 2017. Why Regulate Private Firm Disclosure and Auditing? Accounting and Business Research 47: 473–502.
- Nallareddy, Suresh, Robert Pozen, and Shivaram Rajgopal. 2017. Consequences of Mandatory Quarterly Reporting: The U.K. Experience. Columbia Business School Working Paper No. 17-33.

- Nanda, Ramana, and Matthew Rhodes-Kropf. 2013. Investment Cycles and Startup Innovation. *Journal of Financial Economics* 110: 403–419.
- Porter, Michael E. 1992. Capital Disadvantage: America's Failing Capital Investment System. Harvard Business Review 70: 65–82.
- Romer, Paul M. 1990. Endogenous Technological Change. Journal of Political Economy 98: S71–S102.
- Roychowdhury, Sugata, Nemit Shroff, and Rodrigo Verdi. 2019. The Effects of Financial Reporting and Disclosure on Corporate Investment: A Review. *Journal of Accounting and Economics*, forthcoming.
- Seru, Amit. 2014. Firm Boundaries Matter: Evidence from Conglomerates and R&D Activity. *Journal of Financial Economics* 111: 381–405.
- Shroff, Nemit. 2017. Corporate Investment and Changes in GAAP. *Review of Accounting Studies* 22: 1–63.
- Shroff, Nemit. 2019. Real Effects of PCAOB International Inspections. *The Accounting Review*, forthcoming.
- Shroff, Nemit, Rodrigo S. Verdi, and Benjamin P. Yost. 2017. When Does the Peer Information Environment Matter? *Journal of Accounting and Economics* 64: 183–217.
- Shroff, Nemit, Rodrigo S. Verdi, and Gwen Yu. 2014. Information Environment and the Investment Decisions of Multinational Corporations. *The Accounting Review* 89: 759–790.
- Solomon, Steven D. 2011. In Corporate Disclosure, A Murky Definition of Material. *The New York Times*, April 5.
- Solow, Robert M. 1956. A Contribution to the Theory of Economic Growth. *Quarterly Journal* of Economics 70: 65–94.
- Solow, Robert M. 1957. Technological Change and the Aggregate Production Function. *Review of Economics and Statistics* 39: 312–320.
- Stein, Jeremy C. 1989. Efficient Capital Markets, Inefficient Firms: A Model of Myopic Corporate Behavior. *Quarterly Journal of Economics* 104: 655–669.
- Stoll, John D. 2018. For Companies, It Can Be Hard to Think Long Term. *The Wall Street Journal*, December 3.
- Tian, Xuan, and Tracy Wang. 2014. Tolerance for Failure and Corporate Innovation. *Review* of Financial Studies 27: 211–255.
- Verdi, Rodrigo S. 2012. Discussion of Financial Reporting Frequency, Information Asymmetry, and the Cost of Equity. *Journal of Accounting and Economics* 54: 150–153.
- Yahya, Yasmine. 2016. Much to Debate over Quarterly Reporting. *The Straits Times*, February 2.
- Yiu, Enoch. 2009. Compromise Offered in Quarterly Reporting Row. South China Morning Post, August 3.
- Zhong, Rong I. 2018. Transparency and Firm Innovation. *Journal of Accounting and Economics* 66: 67–93.

| variable definitions | |
|----------------------|----------------------------------------------------------------------------------------------|
| Variable | Definition |
| Measures of innovat | ion |
| PAT | Firm <i>i</i> 's total number of patents filed in year <i>t</i> . |
| TCITE | Firm i's total number of non-self-citations received on the firm's |
| | patents filed in year t. |
| TSM | The total economic value of firm i's patents (based on stock market |
| | reactions to patent grants) filed in year t, expressed in 1982 dollar |
| | values (in million), respectively. |
| Other variables | |
| LNMV | Natural logarithm of firm <i>i</i> 's market value of equity (<u>PRCC_C</u> × <u>CSHO</u>) |
| | measured at the end of fiscal year t. |
| RD | Research and development expenditures (XRD) divided by book value |
| | of total assets (<u>AT</u>) measured at the end of fiscal year t and set to zero |
| | if missing. |
| ROA | Operating income before depreciation (OIBDP) divided by book value |
| | of total assets (AT), measured at the end of fiscal year t . A missing |
| | value is replaced by the industry-year median. |
| PPE | Property, plant, and equipment (net) (PPENT) divided by book value |
| | of total assets (<u>AT</u>) measured at the end of fiscal year t . |
| LEV | Firm <i>i</i> 's leverage ratio, defined as the book value of debt (<u>DLTT+DLC</u>) |
| | divided by book value of total assets (\underline{AT}) measured at the end of fiscal |
| | year t. |
| CAPEX | Capital expenditures (<u>CAPXV</u>) scaled by the book value of total assets |
| | (\underline{AT}) measured at the end of fiscal year t. |
| HERF | Herfindahl index of four-digit SIC industry j to which firm i belongs, |
| | measured at the end of fiscal year t. |
| HERF_SQR | The square of <i>HERF</i> . |
| Q | Firm <i>i</i> 's market-to-book ratio during fiscal year t , calculated as the |
| | market value of equity (<u>PRCC_C</u> * <u>CSHO</u>) plus book value of assets |
| | (\underline{AT}) minus book value of equity (<u>CEQ</u>) minus balance sheet deterred |
| | taxes (set to zero if missing) (<u>TXDB</u>) divided by book value of assets (ATB) |
| | (\underline{AT}) . |
| HPINDEX | $-0.73 \times \log(ASSETS) + 0.043 \times \log(ASSETS)^2 - 0.040 \times AGE$, where |
| | ASSETS is the book value of total assets (<u>A1</u>), and AGE is the number |
| | of years the firm has been on Compustat with a non-missing stock |
| | price. In calculating this index, ASSEIS is replaced with \$4.5 billion |
| | and AGE with thirty-seven years if the actual values exceed these |
| INACE | Unresholds. |
| LNAGE | ivatural logarithm of one plus firm <i>i</i> 's age, approximated by the number |
| | of years listed on Compustat. |
| AMINUD | daily absolute stock rature divided by the dallar terding relieves |
| | (massured in 1000s) |
| | (measured in 1000s). |
| IKEAI | A dummy variable that equals one for treatment firms that experience |
| DEED | an increase in reporting frequency and zero otherwise. |
| PEEK | A dummy variable that equals one for peer firms that do not experience |
| | any change in reporting frequency and zero otherwise. Peer firms are |

Appendix A. Definition of variables Variable definitions

31

| CONTROL | matched to treatment firms based on the closest propensity score and (Fama-French 48) industry.A dummy variable that equals one for control firms that do not experience any change in reporting frequency and zero otherwise. |
|---------------------|------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|
| BEFORE ² | Control firms from industries that have never experienced any change in reporting frequency are matched to treatment firms based on the closest propensity score. A dummy variable that equals one if a firm-year observation is from the second year before the frequency change (year -2) and zero |
| BEFORE ¹ | A dummy variable that equals one if a firm-year observation is from |
| AFTER ⁰ | the year right before the switch year (year -1) and zero otherwise. A dummy variable that equals one if a firm-year observation is from the year of the frequency change (year 0) and zero otherwise |
| AFTER ¹ | A dummy variable that equals one if a firm-year observation is from |
| AFTER ² | the first year after the switch year (year 1) and zero otherwise. A dummy variable that equals one if a firm-year observation is from the second year after the frequency change (year 2) and zero otherwise |
| AFTER ³ | A dummy variable that equals one if a firm-year observation is from |
| AFTER ⁴⁺ | the third year after the frequency change (year 3) and zero otherwise. A dummy variable that equals one if a firm-year observation is from the fourth year or later after the frequency change (year 4+) and zero otherwise |
| QUARTERLY | A dummy variable that equals one for treatment firms that increase reporting frequency to the quarterly level over the sample period and zero otherwise. |
| SEMIANNUAL | A dummy variable that equals one for treatment firms that increase reporting frequency to the semiannual level over the sample period and zero otherwise |
| POST_Q | A dummy variable that equals one if a firm-year observation is from a year after the reporting frequency switch to the quarterly level and zero otherwise |
| POST_S | A dummy variable that equals one if a firm-year observation is from a year after the reporting frequency switch to the semiannual level and zero otherwise. |
| TT1 ' 1' 1 '1 | |

This appendix describes the calculation of variables used in the main analyses of this study.

| Dependent Variable | CHANGE |
|--------------------|-----------|
| LNMV | -0.012 |
| | (-0.45) |
| RD | -3.789** |
| | (-2.10) |
| ROA | -0.202 |
| | (-0.72) |
| PPE | -0.196 |
| | (-1.16) |
| LEV | -0.154 |
| | (-0.92) |
| CAPEX | 1.325*** |
| | (2.95) |
| HINDEX | 0.295 |
| | (0.83) |
| HINDEX_SQR | -0.166 |
| | (-0.55) |
| Q | 0.024 |
| | (0.99) |
| HPINDEX | 0.198*** |
| | (4.00) |
| LNAGE | -0.288*** |
| | (-9.71) |
| AMIHUD | 0.383 |
| | (0.70) |
| PAT_GROWTH | 0.005 |
| | (1.50) |
| TCITE_GROWIH | 0.001 |
| | (1.21) |
| ISM_GROWIH | 0.001 |
| | (0.29) |
| Pseudo R-squared | 0.079 |
| Observations | 9,904 |

Appendix B. Propensity score regression

This table reports the parameter estimates from a probit model used to estimate propensity scores for firm *i*'s change in reporting frequency in year *t*. The dependent variable *CHANGE* is a dummy variable with the value of 1 for increases in reporting frequency and 0 for no change in reporting frequency in year *t*. The two-tailed test *z*-statistics in parentheses are based on standard errors clustered by firm. The sample contains 9,904 firm-year observations from 1951 to 1973. The innovation growth variables, i.e., the growth in the number of patents (*PAT_CHG*), the growth in the number of non-self-citations a firm's patents receive (*TCITE_CHG*), and the growth in the value of a firm's patents (*TSM_CHG*), are computed over prior three-year periods. All other variables are defined in Appendix A. ***, **, and * indicate significance at the 1%, 5%, and 10% levels, respectively.





Note: Patent data are obtained from Kogan et al. (2017). "Total" is calculated as the total number of patents by all entities in a year divided by the total number of patents in 1973. "Public" ("Other") refers to the total number of patents by public firms (other entities) relative to the total number of patents by all entities in a year. "Relative" is calculated as the ratio of the total number of patents by public firms ("Public") to the total number of patents by other entities ("Other").

Figure 2. Trends of innovation proxies around the mandatory increase in reporting frequency



Panel A: The total number of patents







Panel C: The total economic value of patents

Note: "Treat" refers to treatment firms that experience an increase in reporting frequency. "Peer" refers to peer firms that do not experience any change in reporting frequency and are matched to treatment firms based on the closest propensity score and (Fama-French 48) industry. "Control" refers to control firms that are matched to treatment firms based on the closest propensity score but from industries that have never experienced any change in reporting frequency. The values of all innovation proxies are adjusted by sample averages in each year and standardized to range between zero and one.

Table 1

| Event studies | | | | | | | |
|--------------------------------------------------------------------------------------|----------------------|---------------------|------------|--|--|--|--|
| Panel A: SEC announcement on quarterly reporting requirement for all listed firms on | | | | | | | |
| September 15, 1969 | | | | | | | |
| CAR [0, 2] | Semiannual reporters | Quarterly reporters | Difference | | | | |
| Maan | -0.010* | 0.002 | -0.012** | | | | |
| Mean | (-1.75) | (1.21) | (-2.23) | | | | |
| Observations | 80 | 908 | | | | | |

Panel B: President Trump's tweet on dropping quarterly reporting requirement on August 17, 2018

| CAR [0, 2] | Innovative firms | Non-innovative firms | Difference |
|--------------|------------------|-------------------------|------------|
| Mean | 0.006*** | 0.003*** | 0.003** |
| | (5.46) | (8.01) | (2.29) |
| Observations | 1,023 | 6,397 | |

This table reports the 3-day cumulative abnormal stock returns around two events. Panel A reports the results on the SEC announcement of quarterly reporting requirement for all listed firms on September 15, 1969. Its sample includes all firms with non-missing stock returns and reporting frequency. Panel B reports the results on President Trump's tweet on dropping quarterly reporting requirement on August 17, 2018. Its sample includes all listed firms (non-innovative firms) are defined as those with (without) patents filed in the period between 2005 and 2014. CAR [0, 2] is the cumulative market-adjusted abnormal returns during the three-day window starting from the announcement date to two days after the announcement date. The *t*-statistics are reported in parentheses. ***, **, and * indicate significance at the 1%, 5%, and 10% levels, respectively.

| Descriptive statistics | | | | | | |
|------------------------------|-------|-----------|-----------|-----------|-----------|-----------|
| Panel A: Sample distribution | | | | | | |
| Period | Ν | Freq=1(%) | Freq=2(%) | Freq=3(%) | Freq=4(%) | Total (%) |
| 1951-1954 | 501 | 7.78 | 22.55 | 2.99 | 66.67 | 5.06 |
| 1955-1969 | 5,929 | 1.08 | 10.47 | 1.84 | 86.61 | 59.86 |
| 1970-1973 | 3,474 | 0.55 | 1.64 | 1.21 | 96.60 | 35.08 |
| Total | 9,904 | 1.23 | 7.99 | 1.68 | 89.11 | 100.00 |
| No. of unique firms | 1,117 | 58 | 189 | 128 | 1,089 | |

Table 2

Panel B: Descriptive statistics of the sample

| Variable | Ν | Mean | Std Dev | Q1 | Median | Q3 |
|----------|-------|--------|---------|--------|--------|--------|
| PAT | 9,904 | 6.464 | 15.483 | 0.000 | 0.000 | 4.000 |
| TCITE | 9,904 | 46.262 | 108.416 | 0.000 | 0.000 | 28.000 |
| TSM | 9,904 | 4.319 | 14.053 | 0.000 | 0.000 | 0.821 |
| LNMV | 9,904 | 3.830 | 1.641 | 2.623 | 3.697 | 4.972 |
| RD | 9,904 | 0.006 | 0.015 | 0.000 | 0.000 | 0.000 |
| ROA | 9,904 | 0.148 | 0.084 | 0.100 | 0.144 | 0.195 |
| PPE | 9,904 | 0.319 | 0.170 | 0.198 | 0.294 | 0.413 |
| LEV | 9,904 | 0.216 | 0.157 | 0.088 | 0.205 | 0.319 |
| CAPEX | 9,904 | 0.063 | 0.053 | 0.026 | 0.052 | 0.085 |
| HERF | 9,904 | 0.483 | 0.313 | 0.225 | 0.385 | 0.727 |
| Q | 9,904 | 1.703 | 1.042 | 1.055 | 1.422 | 1.967 |
| HPINDEX | 9,904 | -2.365 | 0.693 | -2.887 | -2.430 | -1.943 |
| LNAGE | 9,904 | 1.458 | 0.955 | 0.693 | 1.609 | 2.197 |
| AMIHUD | 9,904 | 0.015 | 0.032 | 0.000 | 0.004 | 0.014 |

This table reports the sample distribution (in Panel A) and the descriptive statistics for the sample (in Panel B). The sample contains 9,904 firm-year observations from 1951 to 1973. The variables are defined in Appendix A.

Table 3

Distribution of the sample for the difference-in-differences tests

| Panel A: Distribution across frequency | | | | | | |
|----------------------------------------|------------------------|-------------------------|-----------------------|-------|--|--|
| Period | Increase to semiannual | Increase to three times | Increase to quarterly | Total | | |
| 1951-1954 | 57 | 20 | 89 | 166 | | |
| 1955-1969 | 252 | 157 | 426 | 835 | | |
| 1970-1973 | 9 | 13 | 52 | 74 | | |
| Total | 318 | 190 | 567 | 1,075 | | |
| No. of firms with non-zero patent | 132 | 85 | 274 | 491 | | |

Panel B: Distribution across switch reason

| Deriod | SEC requirement | Exchange | Investor | Total |
|-------------------|-----------------|-------------|----------|-------|
| I CHOU | SEC requirement | requirement | demand | Total |
| 1951-1954 | 0 | 0 | 166 | 166 |
| 1955-1969 | 305 | 133 | 397 | 835 |
| 1970-1973 | 61 | 0 | 13 | 74 |
| Total | 366 | 133 | 576 | 1,075 |
| No. of firms with | 144 | 51 | 206 | 401 |
| non-zero patent | 144 | 51 | 290 | 491 |

This table reports the distribution of the sample for the difference-in-differences tests. Panel A reports the time-series distribution of treatment firms experiencing an increase in reporting frequency across reporting frequency and Panel B presents such distribution across switch reason. The distribution of treatment firms with a non-zero patent is reported at the end of each panel.

Table 4

Simple difference-in-differences test

| Variable | PAT | TCITE | TSM | | | |
|-------------------------------------------------------------|-------------------|----------------|-----------|--|--|--|
| variable | (1) | (2) | (3) | | | |
| Panel A: Mandatory increases in reporting frequency (N=499) | | | | | | |
| (1) Mean TREAT difference | -1.071*** | -12.404*** | -1.020*** | | | |
| (After – Before) | (-10.71) | (-22.59) | (-20.46) | | | |
| (2) Mean PEER difference | 0.137 | 8.935*** | 0.981*** | | | |
| (After – Before) | (0.48) | (3.33) | (3.66) | | | |
| (3) Mean CONTROL difference | 0.794 | 7.179 | 0.739 | | | |
| (After – Before) | (1.65) | (0.96) | (1.54) | | | |
| (4) Mean DiD estimator | -1.865*** | -19.583** | -1.758*** | | | |
| $(\Delta TREAT - \Delta CONTROL)$ | (-3.79) | (-2.61) | (-3.64) | | | |
| (5) Mean DiD estimator | -0.657 | 1.756 | 0.242 | | | |
| $(\Delta PEER - \Delta CONTROL)$ | (-1.18) | (0.22) | (0.40) | | | |
| (6) Mean DiD estimator | -1.208*** | -21.339*** | -2.000*** | | | |
| $(\Delta TREAT - \Delta PEER)$ | (-4.01) | (-7.78) | (-7.33) | | | |
| | | | | | | |
| Panel B: Voluntary increases in rep | porting frequency | <u>(N=576)</u> | | | | |
| (1) Mean TREAT difference | -0.738*** | -5.688*** | -0.809*** | | | |
| (After – Before) | (-4.08) | (-7.27) | (-7.64) | | | |
| (2) Mean PEER difference | 1.260*** | 11.159*** | 1.207** | | | |
| (After – Before) | (5.00) | (4.74) | (5.33) | | | |
| (3) Mean CONTROL difference | 1.417* | 6.561 | 1.444* | | | |
| (After – Before) | (1.88) | (1.31) | (1.77) | | | |
| (4) Mean DiD estimator | -2.155*** | -12.249** | -2.253*** | | | |
| $(\Delta TREAT - \Delta CONTROL)$ | (-2.78) | (-2.42) | (-2.74) | | | |
| (5) Mean DiD estimator | -0.157 | 4.598 | -0.237 | | | |
| $(\Delta PEER - \Delta CONTROL)$ | (-0.20) | (0.89) | (-0.28) | | | |
| (6) Mean DiD estimator | -1.998*** | -16.848*** | -2.016*** | | | |
| $(\Delta TREAT - \Delta PEER)$ | (-6.44) | (-6.79) | (-8.06) | | | |

This table provides the univariate difference-in-differences test results. Panel A (B) presents the results for mandatory (voluntary) increases in reporting frequency. *PAT* (*TCITE*, or *TSM*) is the sum of firm *i*'s number of patents (number of citations, or value of patents), adjusted by sample averages in each year, over the three-year window before or after the frequency switch. *TREAT*, *PEER*, and *CONTROL* refer to firms experiencing increases in reporting frequency, the matched peer firms from the same industry, and the matched control firms from other industries. All variables are defined in Appendix A. The *t*-statistics are given in parentheses below the mean differences in innovation outcomes. ***, **, and * indicate significance at the 1%, 5%, and 10% levels, respectively.

| Dependent variable | PAT | TCITE | TSM |
|------------------------------|---------------------------|------------------|---------------|
| | (1) | (2) | (3) |
| Panel A: Mandatory increase | es in reporting frequency | | |
| TREAT×BEFORE ² | -0.181 | -4.927 | -0.190 |
| | (-0.31) | (-1.17) | (-0.95) |
| TREAT×BEFORE ¹ | -0.261 | -1.893 | -0.156 |
| | (-0.34) | (-0.49) | (-0.34) |
| $\Gamma REAT 	imes AFTER^1$ | -1.315** | -18.857** | -0.863** |
| | (-2.12) | (-1.97) | (-2.18) |
| TREAT×AFTER ² | -1.939* | -18.935 | -1.701** |
| | (-1.82) | (-1.27) | (-2.13) |
| TREAT×AFTER ³ | -3.554*** | -33.455* | -3.262** |
| | (-2.81) | (-1.71) | (-2.19) |
| PEER×BEFORE ² | 0.079 | -3.540 | -0.058 |
| | (0.13) | (-0.92) | (-0.26) |
| $PEER \times BEFORE^1$ | 0.824 | -0.842 | -0.040 |
| | (1.01) | (-0.20) | (-0.09) |
| $PEER \times AFTER^1$ | -0.862 | -3.260 | 0.401 |
| | (-1.34) | (-0.40) | (0.99) |
| PEER×AFTER ² | 0.293 | 0.297 | 0.143 |
| | (0.28) | (0.02) | (0.19) |
| $PEER \times AFTER^3$ | -0.953 | 0.387 | 0.065 |
| | (-0.88) | (0.02) | (0.05) |
| BEFORE ² | 0.076 | 5.859 | 0.293 |
| | (0.13) | (1.41) | (1.54) |
| SEFORE ¹ | 1.141 | 4.277 | 0.409 |
| | (1.65) | (1.04) | (0.83) |
| $AFTER^1$ | 1 225* | 21 307** | 0 894** |
| | (1.92) | (2.16) | (2,00) |
| $FTER^2$ | 2 045* | 20 273 | 1 715** |
| | (1.96) | (1.32) | (2 14) |
| FTER ³ | 2 995** | 30 554 | 3 162** |
| | (2.47) | (1.58) | (2.17) |
| irm FE | (2.77) Ves | (1.50) Yes | (2.17) Ves |
| /ear FF | Vec | Vec | Ves |
| diusted R-squared | 0 725 | 0 7/1 | 0 7/8 |
| bservations | 0.725 A 837 | 4 837 | 1 827 |
| 10501 Valions | 4,032 | т,032 | +,052 |
| Panel B: Voluntary increases | s in reporting frequency | | |
| TREAT×BEFORE ² | 0.351 | -9.197 | -0.220 |
| | (0.78) | (-1.50) | (-0.73) |
| TREAT×BEFORE ¹ | -0.047 | -7.088 | -0.387 |
| | (-0.09) | (-1.19) | (-0.89) |
| $TREAT \times AFTER^{1}$ | -2.375* | -16.084** | -2.178* |
| | (-1.94) | (-1.97) | (-1.90) |
| REAT×AFTER ² | -2.835** | -18 906 | -2 796* |
| | (-2.03) | (-1.43) | (-1.91) |
| | (-2.05) | (-1, - 5) | (-1,71) |

Table 5Difference-in-differences regression analyses

| TREAT×AFTER ³ | -2.518** | -23.967* | -3.503** |
|--------------------------|----------|----------|----------|
| | (-2.18) | (-1.88) | (-2.24) |
| PEER×BEFORE ² | -0.341 | -9.496 | -0.356 |
| | (-0.88) | (-1.56) | (-1.22) |
| $PEER \times BEFORE^1$ | -0.087 | -5.943 | -0.437 |
| | (-0.17) | (-0.99) | (-1.00) |
| $PEER \times AFTER^1$ | -1.375 | 0.468 | -1.396 |
| | (-1.13) | (0.05) | (-1.22) |
| $PEER \times AFTER^2$ | -0.140 | -1.010 | -0.648 |
| | (-0.10) | (-0.08) | (-0.44) |
| $PEER \times AFTER^3$ | -0.061 | 0.840 | 0.104 |
| | (-0.05) | (0.07) | (0.06) |
| BEFORE ² | 0.358 | 9.825* | 0.365 |
| | (0.95) | (1.69) | (1.32) |
| BEFORE ¹ | 0.810** | 8.159 | 0.656* |
| | (1.96) | (1.41) | (1.73) |
| AFTER ¹ | 2.256* | 16.893** | 1.817* |
| | (1.90) | (2.34) | (1.70) |
| AFTER ² | 2.715** | 19.890 | 2.464* |
| | (2.04) | (1.64) | (1.85) |
| AFTER ³ | 2.351** | 24.666** | 3.041** |
| | (2.22) | (2.15) | (2.16) |
| Firm FE | Yes | Yes | Yes |
| Year FE | Yes | Yes | Yes |
| Adjusted R-squared | 0.580 | 0.633 | 0.453 |
| Observations | 5,668 | 5,668 | 5,668 |

This table reports regression estimates of the innovation dynamics of treatment and control firms surrounding the frequency switch, i.e., three years before and after the frequency change. Panel A (B) presents the results for mandatory (voluntary) increases in reporting frequency. The dependent variable is innovation output measured by *PAT*, *TCITE*, or *TSM* in a given year. *TREAT* and *PEER* refer to treatment firms that experience increases in reporting frequency and the matched peer firms from the same industry. Firm and year fixed effects are included in all regressions. All variables are defined in Appendix A. The two-tailed test *t*-statistics in parentheses are based on standard errors clustered by firm. ***, **, and * indicate significance at the 1%, 5%, and 10% levels, respectively.

| | undryses bused on the r | un sumpte | |
|--------------------|-------------------------|-------------|------------|
| Danandant yariahla | PAT | TCITE | TSM |
| Dependent variable | (1) | (2) | (3) |
| QUARTERLY | -5.438*** | -25.541*** | -5.151*** |
| ×POST | (-6.35) | (-5.31) | (-5.24) |
| SEMIANNUAL | -2.187 | -7.815 | -1.194 |
| ×POST | (-1.40) | (-0.86) | (-1.31) |
| LNMV | 1.979*** | 18.465*** | 3.675*** |
| | (3.66) | (4.85) | (5.06) |
| RD | 156.208*** | 1033.758*** | 154.557*** |
| | (4.84) | (5.59) | (4.21) |
| ROA | -5.268* | -40.125** | -8.815** |
| | (-1.77) | (-2.17) | (-2.37) |
| PPE | 2.234 | 15.626 | 4.506 |
| | (0.85) | (0.84) | (1.29) |
| LEV | 1.300 | 6.841 | 0.017 |
| | (0.72) | (0.62) | (0.01) |
| CAPEX | -13.560*** | -79.925*** | -16.439*** |
| | (-4.05) | (-3.44) | (-3.48) |
| HERF | 0.289 | 2.377 | 0.611 |
| | (0.04) | (0.05) | (0.08) |
| HERF_SQR | 0.604 | 6.607 | 1.566 |
| | (0.11) | (0.18) | (0.25) |
| Q | -0.709** | -3.369* | -0.894** |
| | (-2.18) | (-1.77) | (-1.98) |
| HPINDEX | 0.693* | 1.244 | 1.273*** |
| | (1.93) | (0.50) | (2.71) |
| LNAGE | -2.289*** | -11.413*** | -3.490*** |
| | (-3.16) | (-2.59) | (-4.30) |
| AMIHUD | 7.851*** | 40.680*** | 12.730*** |
| | (4.08) | (3.22) | (5.66) |
| Firm FE | Yes | Yes | Yes |
| Year FE | Yes | Yes | Yes |
| Adjusted R-squared | 0.729 | 0.761 | 0.643 |
| Observations | 9,904 | 9,904 | 9,904 |

 Table 6

 Difference-in-differences analyses based on the full sample

This table reports regression estimates of the impact of reporting frequency on innovation based on the full sample. The dependent variable is innovation output measured by *PAT*, *TCITE*, or *TSM* in a given year. *QUARTERLY* (*SEMIANNUAL*) refers to treatment firms that increase reporting frequency to the quarterly (semiannual) level. Firm and year fixed effects are included in all regressions. All variables are defined in Appendix A. The two-tailed test *t*-statistics in parentheses are based on standard errors clustered by firm. ***, **, and * indicate significance at the 0.01, 0.05, and 0.10 levels, respectively.

| D finance changes in milovation output based on the run sample | | | | | |
|----------------------------------------------------------------|-----------|-----------|-----------|--|--|
| Dependent variable | PAT | TCITE | TSM | | |
| | (1) | (2) | (3) | | |
| QUARTERLY | 1.199 | -1.415 | 0.040 | | |
| $\times BEFORE^2$ | (1.46) | (-0.31) | (0.06) | | |
| QUARTERLY | 0.531 | -1.919 | -0.567 | | |
| $\times BEFORE^{1}$ | (0.60) | (-0.36) | (-0.72) | | |
| QUARTERLY | -1.750** | -5.262 | -1.723** | | |
| $\times AFTER^0$ | (-2.04) | (-1.10) | (-1.97) | | |
| QUARTERLY | -1.687* | -6.384 | -1.772* | | |
| $\times AFTER^1$ | (-1.91) | (-1.32) | (-1.94) | | |
| QUARTERLY | -1.733* | -7.541 | -1.870* | | |
| $\times AFTER^2$ | (-1.88) | (-1.37) | (-1.88) | | |
| QUARTERLY | -2.916*** | -9.703* | -2.529*** | | |
| $\times AFTER^3$ | (-3.15) | (-1.71) | (-2.62) | | |
| QUARTERLY | -3.647*** | -17.009** | -4.179*** | | |
| $\times AFTER^{4+}$ | (-3.10) | (-2.35) | (-3.12) | | |
| Controls | Yes | Yes | Yes | | |
| Firm FE | Yes | Yes | Yes | | |
| Year FE | Yes | Yes | Yes | | |
| Industry-specific trend | Yes | Yes | Yes | | |
| Adjusted R-squared | 0.761 | 0.788 | 0.697 | | |
| Observations | 9,904 | 9,904 | 9,904 | | |

 Table 7

 Dynamic changes in innovation output based on the full sample

This table reports regression estimates of the innovation dynamics surrounding the frequency switch based on the full sample. The dependent variable is innovation output measured by *PAT*, *TCITE*, or *TSM* in a given year. *QUARTERLY* (*SEMIANNUAL*) refers to treatment firms that increase reporting frequency to the quarterly (semiannual) level. All regressions include the same set of control variables as in Table 6, *SEMIANNUAL* interacted with leads and lags, firm and year fixed effects, and industry-specific linear trends. All variables are defined in Appendix A. The two-tailed test *t*-statistics in parentheses are based on standard errors clustered by firm. ***, **, and * indicate significance at the 0.01, 0.05, and 0.10 levels, respectively.